Author’s response to reviews

Title: Rare cause of severe anemia due to pyogenic granuloma in the jejunum

Authors:
Shun-ichi Misawa (96084sm@jichi.ac.jp)
Hiroto Sakamoto (m05041hs@jichi.ac.jp)
Akira Kurogochi (159735@hp-hata.com)
Yasushi Kirii (yasukiri@jichi.ac.jp)
Shinichiro Nakamura (139882@hp-hata.com)
Tomoko Misawa (96014to@jichi.ac.jp)
Suguru Yoneda (119865@hp-hata.com)
Mari Hirano (159905@hp-hata.com)
Osamu Owa (0809@hp-hata.com)
Hiroyuki Takagi (0344@hp-hata.com)
Hiroyoshi Ota (hohta@shinshu-u.ac.jp)

Version: 2 Date: 11 Sep 2015

Author’s response to reviews

September 11 2015
Dear Elaine Zhang, BioMed central
BMC Gastroenterology

Dear Editor:

I appreciate that you decided to accept our manuscript titled “Rare cause of severe anemia due to pyogenic granuloma in the jejunum,”. I made list of abbreviations after the conclusion section and compose FIGURE 4a-c into a single image and labeled them. We believe that the findings of this study are relevant to the scope of your journal and will be of interest to its readership.

We look forward to hearing from you at your earliest convenience.
Sincerely,

Shun-ichi Misawa
Matsumoto City Hospital
3901401 4417-180 Hata Matsumoto Nagano Japan
Phone No:0263-92-3027
Email Address: 96084sm@jichi.ac.jp